Author's response to reviews

Title: Feasibility and Validity of On-The-Phone Screening of Adverse Events after Voluntary Medical Male Circumcision in Swaziland

Authors:

Tigistu Adamu Ashengo (tadamu@jhpiego.net)
Jonathan Grund (iha3@cdc.gov)
Thabo Hlophe (thabo.hlophe@gmail.com)
Munamato Mirira (miriraM@state.gov)
Naomi Bock (neb2@cdc.gov)
Emmanuel Njeuhmeli (enjeuhmeli@usaid.gov)
George Bicego (gpb4@cdc.gov)
Kelly Curran (kcurran@jhpiego.net)
Masitsela Mhalanga (masitsela@epr.co.sz)
Rhoy Shoshore (rshoshore@yahoo.co.uk)
Khumbulani Moyo (khumbulani@psi.sz)
Elizabeth Mallas (lizmallas@gmail.com)
Laura Fitzgerald (lfitzgerald@jhpiego.net)

Version: 2 Date: 1 July 2013

Author's response to reviews:

Correction on Authors order is made.